Author’s response to reviews

Title: Prevalence of potential drug-drug interactions and associated factors among outpatients and inpatients in Ethiopian Hospitals: a systematic review and Meta-analysis of observational studies

Authors:

Wondim Ayenew (yimesgen20@gmail.com)

Getahun Asmamaw (getahunasmamaw12@gmail.com)

Arebu Issa (issaareb@gmail.com)

Version: 1 Date: 15 May 2020

Author’s response to reviews:

Responses to reviewers
We revised the manuscript based on the reviewers’ comments.
For reviewer 1:
we modified the manuscript based on the suggested comment. We said sufficient data when the article fulfills the inclusion and exclusion criteria and contains the outcome of interest based on our data collection checklist.
Reviewr 2:
We revised our manuscript based on the given comments. The role of nutrients in the development of DDIs did not discussed because it was not our objective and mentioned it in exclusion criteria. The time of DDIs development was not discussed because none of the included articles mentioned it.
Reviewer 3:
We revised our manuscript based on the given comments. You mentioned, RISMA is used to report reviews evaluating randomized trials, but can also be used as a basis for reporting systematic reviews of other types of research, particularly evaluations of interventions. We already registered the protocol on PRISMA and does not used MOOSE.
Reviewer 4:
We revised our manuscript based on the given comments. We add ert fatigue. We mentioned the effect of interaction as risk of interaction.